Author’s response to reviews

Title: Multicentric, multifocal, and recurrent osteoid osteoma of the hip: first case report

Authors:

Maria Cristina Cortese (cristinacortese1010@gmail.com)
Domenico Albano (albanodomenico@me.com)
Carmelo Messina (carmelomessina.md@gmail.com)
Giuseppe Perrucchini (pergiudoc@yahoo.com)
Enrico Gallazzi (enrico.gallazzi@gmail.com)
Mauro Battista Gallazzi (maurobattista.gallazzi@gpini.it)
Primo Andrea Daolio (primoandrea.daolio@gpini.it)
Luca Maria Sconfienza (io@lucasconfienza.it)

Version: 4 Date: 27 Mar 2019

Author’s response to reviews:

Editor Comments:

1. Please combine your discussion and conclusions under one heading "Discussion and Conclusions".

Answer: Done.

2. In the Availability of data and materials section please clearly state who should be contacted if someone wants to request the data.

Answer: Done.

3. Thank you for providing your CARE checklist. However, at this stage it is not required and so we kindly ask that you remove it from your manuscript.

Answer: We have removed the Care checklist.